Author's response to reviews

Title: The effectiveness of combined bracing and exercise in adolescent idiopathic scoliosis based on SRS and SOSORT criteria: A prospective study.

Authors:

Stefano Negrini (stefano.negrini@med.unibs.it)
Sabrina Donzelli (sabrina.donzelli@isico.it)
Monia Lusini (monia.lusini@isico.it)
Salvatore Minnella (salvatore.minnella@isico.it)
Fabio Zaina (fabio.zaina@isico.it)

Version: 5 Date: 5 June 2014

Author's response to reviews:

1. Title page formatted as required
2. Abstract changed and shortened to 350 words
3. Format of the paper according to BMC requirements